Author's response to reviews

Title: Liver and brain abscess caused by Aggregatibacter paraphrophilus in association with a large patent foramen ovale: a case report

Authors:

Shaumya Ariyaratnam (shaumyariyaratnam@gmail.com)
Parag R Gajendragadkar (gajendragadkar@gmail.com)
Richard J Dickinson (richard.dickinson@hinchingbrooke.nhs.uk)
Phil Roberts (Phil.RobertsGastro@hinchingbrooke.nhs.uk)
Kathryn Harris (HarriK2@gosh.nhs.uk)
Andrew Carmichael (Andrew.carmichael@addenbrookes.nhs.uk)
Johannis A Karas (andreas.karas@papworth.nhs.uk)

Version: 3 Date: 22 January 2010

Author's response to reviews:

Dear Editors, we have made the changes to the proof as requested and corrected typographical errors as detailed below:

1) at the bottom of page 5, "Results of the 16S rRNA PCR of the brain biopsy again detected ......." Changed to "16S rDNA PCR" instead of "16S rRNA PCR";

2) Amended the list of authors in the manuscript to the preferred style is 'First name, Initial, Last name' (e.g. Joe F Bloggs) as requested by JMCR.

3) Changed the formatting of the cited references in the text to square brackets instead of superscript as requested by JMCR.

4) Removed the authors' qualifications and job titles from the manuscript as requested by JMCR.

5) Author affiliation addresses out in full as requested by JMCR.

6) On page 4, first paragraph to "may have been accidentally swallowed" as this was omitted.

7) Typographical errors:
On page 5 3rd paragraph: "apyrexial" to "pyrexial"
On page 4 last paragraph, added "diameter" after 12mm as was omitted.

8) Formatted the references to conform to the JMCR regulations;

yours sincerely
Andreas Karas